Author's response to reviews

Title: A novel de novo mutation in the serine-threonine kinase STK11 gene in a Korean patient with Peutz-Jeghers syndrome

Authors:

Jong-Ha Yoo (jhyoo92@nhimc.or.kr)
Jee-Hyoung Yoo (jhlovys@nhimc.or.kr)
Yoon-Jung Choi (chris316@nhimc.or.kr)
Jung-Gu Kang (kangski@nhimc.or.kr)
Young-Kyu Sun (labsun@nhimc.or.kr)
Chang-Seok Ki (changski@skku.edu)
Kyung-A Lee (KAL1119@yuhs.ac)
Jong-Rak Choi (cj0606@yuhs.ac)

Version: 4 Date: 19 May 2008

Author's response to reviews: see over
2008-5-19

Dear Editor,

We would like to resubmit the manuscript entitled, “A novel de novo mutation in the serine-threonine kinase STK11 gene in a Korean patient with Peutz-Jeghers syndrome”, by Yoo et al., for consideration for possible publication in the BMC Medical Genetics.

<Point-by-point response to editor’s concerns>

1. Author list check: We corrected the list of authors in the submission system.
2. Abbreviations: We presented these as a paragraph rather than a list.
3. Re-title ‘Figures’; We re-titled ‘Figure legends’

The data shown here have not been submitted for publication nor has it been published in whole or in part elsewhere. All the authors took part in the work and agree with the contents of the manuscript consent to its submission to BMC Medical Genetics. The category “Research article” seems most appropriate for our manuscript. It would be a great honor to have an article accepted in the BMC Medical Genetics. We would greatly appreciate being considered for publishing our report and hope to hear good news from you soon. Thank you very much.

Sincerely yours,

Jong-Rak Choi, M.D., Ph.D.
Department of Laboratory Medicine
Yonsei University College of Medicine
250 Seongsanno, Seodaemun-gu
Seoul 120-752, Korea
E-mail: cjro606@yuhs.ac
Tel: +82-02-2228-2445
Fax: +82-02-313-0956

Authors:
Jong-Ha Yoo: jhyoo92@nhimc.or.kr
Jee-Hyoung Yoo: jhlovys@nhimc.or.kr
Yoon-Jung Choi: chris316@nhimc.or.kr
Jung-Gu Kang: kangski@nhimc.or.kr
Young-Kyu Sun: labsun@nhimc.or.kr
Chang-Seok Ki: changski@skku.edu
Kyung-A Lee: KAL1119@yuhs.ac